Dear editors and reviewers

Re: Manuscript ID: 79387 and Title: Prognostic analysis of patients with combined hepatocellular-cholangiocarcinoma after radical resection: A retrospective multicenter study

Thank you for your precious comments and advice. Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. We have studied comments carefully and have made correction which we hope meet with approval. The main corrections in the paper and the responds to the reviewer’s comments are as flowing:

Response to the reviewer’s comments:

Reviewer #1:

Q1. Please remove the supplementary to another document.

Response: We agree with the comment. Supplementary information was removed to another document from manuscript and supplementary document was uploaded additionally.

Q2. Manuscript should be edited, and updated according to the journal’s guidelines.

Response: We are grateful for the suggestion. We revisited the journal’s guidelines and updated our manuscript.

Thank you for your careful review.

Reviewer #2:

We appreciate your comments on our manuscript and have made changes based on the comments of other reviewers and editors. Thank you for your careful review.

Reviewer #3:

Q1. There are some minor language polishings, which require a minor editing.

Response: Thank you for your careful review. We are very sorry for the mistakes in this manuscript and inconvenience they caused in your reading. The manuscript has been thoroughly revised and rewritten by American Journal Experts (AJE) so we hope it can meet the journal’s standard.

Q2. The figures are somewhat small, please make an update to the images.
Response: We are sorry for the inconvenience that the small images caused to your reading. We have resubmitted larger images in a separate file to make it easier for editors and reviewers to read.

Q3. References list should be edited with PMID.

Response: We apologize for the formatting issues in the original manuscript. In the revised manuscript, we added PMID and DOI for most of the references, however, some of them were published too early to obtain PMID or DOI.

Thank you for your careful review.